Author's response to reviews

Title: Pre-Hospital Treatment of Acute Poisonings in Oslo

Authors:

Fridtjof Heyerdahl (fridtjof.heyerdahl@medisin.uio.no)
Knut E Hovda (kehovda@yahoo.no)
Mari A Bjornaas (mabjornaas@gmail.com)
Anne K Nore (annekathrinenore@yahoo.com)
Jose CP Figueiredo (josf@uus.no)
Oivind Ekeberg (oivind.ekeberg@uus.no)
Dag Jacobsen (daja@uus.no)

Version: 4 Date: 30 October 2008

Author's response to reviews: see over
Dear Dr Jo Appleford

Thank you for the acceptance-in-principle. We have changed the manuscript according to the comments from Paul Dietze, and have gone through the manuscript checklist once more.

Sincerely yours,

Fridtjof Heyerdahl, MD
Department of Acute Medicine
Ullevaal University Hospital
Oslo, Norway